Author’s response to reviews

Title: Controversial Ebola Vaccine Trials in Ghana: A thematic analysis of critiques and rebuttals in digital news

Authors:

Per Egil Kummervold (per.egil.kummervold@norut.no)

William Schulz (william.schulz@lshtm.ac.uk)

Elisabeth Smout (elizabeth.smout@lshtm.ac.uk)

Luis Fernandez-Luque (luis.tromso@gmail.com;luis.luque@norut.no)

Heidi Larson (heidi.larson@lshtm.ac.uk)

Version: 1 Date: 24 Nov 2016

Author’s response to reviews:

Dear Editors,

We have fixed the manuscript, by citing Figure 7, Appendix 2 and 3. We also added the attachment of Appendix 4.

Please let me know if there is anything else we need to do.

Regards,

Luis Fernandez-Luque